Progress Slow for Online Clinical Trial Registries

BY CARL SHERMAN

Contributing Writer

NEW YORK — When questions were raised about possible concealment of clinical trial data, two pharmaceutical companies agreed last year to set up Web sites where such data would be posted.

It appeared at that time that others in the industry would follow suit, "but as it turns out, very little has happened," Norman Sussman, M.D., said at a meeting on

psychopharmacology sponsored by New York University.

An Internet search performed at the beginning of March, followed by inquiries to the companies themselves, found that information was for the most part incomplete, difficult to use, or entirely absent.

"This says something about the goodwill of the companies," said Dr. Sussman, professor of psychiatry at the university.

In June 2004, New York State Attorney

General Eliot Spitzer filed suit against GlaxoSmithKline Inc., charging that the company's selective release of trial data on the use of paroxetine (Paxil) in children constituted consumer fraud. As part of a settlement of the lawsuit at the end of August, the company agreed to post clinical trial results for all GSK drugs on

An inquiry by the attorney general's office into data relating to off-label use of drugs manufactured by Forest Laboratories Inc. led to a similar agreement with that company.

Dr. Sussman's Internet investigation found that one pharmaceutical company has done what was promised, and it was neither of those originally involved: Eli

Its Web site (www.lillytrials.com) supplies clinical trial data on all its psychotropic drugs, broken down into "completed" and "initiated" trials.

For completed trials, the site supplies PDF files of basic information—"not everything you want to know, but a sense of how the study was designed, the method, and outcomes," he said.

For the most part, the information that is posted by the company on the site is raw data: "If you were expecting some-

Information was for the most part incomplete, difficult to use, or entirely absent. 'This says something about the goodwill of

the companies.'

thing simple, it's not here. You have to have an understanding of research methodology to evaluate these," he said.

The "initiated trials" section lists phase 2, 3, and 4 studies that were begun since July 2004, most

of which are still recruiting patients. "The idea is that once you do this, you can no longer hide the results of the study," Dr. Sussman said.

The speed with which Lilly put such complete data on its Web site "tells you that any company that wanted to could do it tomorrow. They all have internal documents that summarize studies," he said.

The GlaxoSmithKline registry (http:// ctr.gsk.co.uk/welcome.asp) is "a little more restricted," he said. The site posts only results of studies that were done after the merger of Glaxo and SmithKline which excludes most of the investigations involving Paxil and children, Dr. Sussman

The GSK presentation includes less narrative discussion of study findings than the Lilly site. "It's mostly numbers. ... You have to look into the statistics and form your own conclusion. It's not intended for the average practitioner," he said.

The other company that agreed to post data, Forest Laboratories, has set up a registry (www.forestclinicaltrials.com), but even though completed trials are listed in the registry, there are as yet no data that are included from ongoing studies. "If you call Forest, they send you from one department to another," Dr. Sussman said.

An industry association, the Pharmaceutical Research and Manufacturers of America, maintains a Web site of its own (www.clinicalstudyresults.org), but clinical trial data on the site, which are supplied by pharmaceutical companies, are fragmentary.

A government site (www.clinicaltrials.gov) lists some ongoing trials but no results, he said.

BRIEF SUMMARY Consult Package insert for full prescribing information

HUMIRA®

(adalimumab)

 $m R\!\!\!\!/$ only

Cases of tuberculosis (frequently disseminated or extrapulmonary at clinical presentation) have been observed in patients receiving HUMIRA. Patients should be evaluated for Italent tuberculosis infection with a tuberculin skin test. Treatment of latent tuberculosis infection should be initiated prior to therapy with HUMIRA.

CONTRAINDICATIONS

HUMIRA should not be administered to patients with known hypersensitivity to HUMIRA or any of its components.

SERIOUS INFECTIONS

SERIOUS INFECTIONS
SERIOUS INFECTIONS AND SEPSIS, INCLUDING FATALITIES, HAVE BEEN REPORTED WITH THE USE OF THE BLOCKING AGENTS INCLUDING HUMIRA. MANY OF THE SERIOUS INFECTIONS HAVE OCCURRED IN PATIENTS ON CONCOMITANT IMMUNOSUPPRESSIVE THERAPY THAT, IN ADDITION TO THEIR RHEUMATOID ARTHRITIS, COULD PREDISPOSE THEM TO INFECTIONS TUBERCULOSIS AND INVASIVE OPPORTUNISTIC FUNGAL INFECTIONS HAVE BEEN OBSERVED IN PATIENTS TREATED WITH THE BLOCKING AGENTS INCLUDING HUMIRA.

INCLUDING HUMIRA.

TREATMENT WITH HUMIRA SHOULD NOT BE INITIATED IN PATIENTS WITH ACTIVE INFECTIONS INCLUDING CHRONIC OR LOCALIZED INFECTIONS. PATIENTS WHO DEVELOP A NEW INFECTION WHILE UNDERGOING TREATMENT WITH HUMIRA SHOULD BE MONITORDED CLOSELY, ADMINISTRATION OF HUMIRA SHOULD BE DISCONTINUED IF A PATIENT DEVELOPS A SERIOUS INFECTION. PHYSICIANS SHOULD EXERCISE CAUTION WHEN CONSIDERING THE USE OF HUMIRA IN PATIENTS WITH A HISTORY OF RECURRENT INFECTION ON PATIENTS WHO HAVE RESIDED IN REGIONS WHERE TUBER-CULOSIS AND HISTOPLASMOSIS ARE ENDEMIC (SEE PRECAUTIONS - TUBER-CULOSIS AND HISTOPLASMOSIS AND HISTOPLASMOSIS ARE ENDEMIC (SEE PRECAUTIONS - TUBER-CULOSIS AND HISTOPLASMOSIS AND HISTOPLASMOSIS AND HISTOPLASMOSIS ARE ENDEMIC (SEE PRECAUTIONS - TUBER-CULOSIS AND HISTOPLASMOSIS AND HISTOPLASMOSIS AND HISTOPLASMOSIS ARE ENDEMIC (SEE PRECAUTIONS - TUBER-CULOSIS AND HISTOPLASMOSIS AND

Use with Anakinra

Serious infections were seen in clinical studies with concurrent use of anakinra (an interleukin-1 antagonist) and another TNF-blocking agent, with no added benefit. Because of the nature of the adverse events seen with this combination therapy, similar toxicities may also result from combination of anakinra and other TNF-blocking agents. Therefore, the combination of HUMIRA anakinra is not recommended (see PRECAUTIONS, Drug Interactions).

Neurologic Events: Use of TNF blocking agents, including HUMIRA, has been associated with rare cases of exacerbation of clinical symptoms and/or radiographic evidence of demyelinating disease. Prescribers should exercise caution in considering the use of HUMIRA in patients with preexisting or recent-onset central nervous system demyelinating disorders.

Malignancies: In the controlled portions of clinical trials of all the TNF-blocking agents, more cases of lymphoma have been observed among patients receiving TNF blockers compared to control patients. During the controlled portions of HUMIRA trials in patients with moderately to severely active rheumatoid arthritis. 2 lymphomas were observed among 1380 HUMIRA-treated patients versus 0 among 690 control patients (mean duration of controlled treatment approximately 7 montrits). In the controlled and open-label portions of these clinical trials of HUMIRA in rheumatoid arthritis patients, over 4870 patient-years of therapy. This is approximately 5-fold higher than expected in the general population. Rates in clinical trials for HUMIRA cannot be compared to rates of clinical trials of other TNF blockers and may not predict the rates observed in a broader patient population. Patients with rheumatoid arthritis, particularly those with highly active disease, are at a higher risk for the development of ymphoma. The potential role of TNF blocking therapy in the development of ymphoma. The potential role of TNF blocking therapy in the development of ymghoma. The potential role of TNF blocking therapy in the developme

nancies).

Hypersensitivity Reactions: In postmarketing experience, anaphylaxis has been reported rarely following HUMIRA administration. If an anaphylactic or other serious allergic reaction occurs, administration of HUMIRA should be discontinued immediately and appropriate therapy instituted. In clinical trials of HUMIRA, allergic reactions overall (e.g., allergic rash, anaphylactioid reaction, fixed drug reaction, non-specified drug reaction, urticaria) have been observed in approximately 1% of nations.

reaction, non-specified drug reaction, urguaria) have been accounted by the foliation and the first state of pancytopenia including aplastic anemia have been reported with TNF blocking agents. Adverse events of the hematologic system, including medically significant cytopenia (e.g. thrombocytopenia, leukopenia) have been infrequently reported with HUMIRA (see ADVERSE REACTIONS, Other Adverse Reactions). The causal relationship of these reports to HUMIRA remains unclear. All patients should be advised to seek immediate medical attention if they develop signs and symptoms suggestive of blood dyscrasias or infection (e.g. persistent fever, bruising, bleeding, pallor) while on HUMIRA. Discontinuation of HUMIRA therapy should be considered in patients with confirmed significant hematologic abnormalities.

Discontinuation of HUMIRA therapy should be considered in patients with confirmed significant hematologic abnormalities.

PRECAUTIONS
Information to Patients: The first injection should be performed under the supervision of a qualified health care professional. If a patient or caregiver is to administer HUMIRA, he/she should be instructed in injection techniques and their ability to inject subcutaneously should be assessed to ensure the proper administration of HUMIRA (see HUMIRA, PATIENT INFORMATION LEAFLET). A puncture-resistant container for disposal of needles and syringes should be used. Patients or caregivers should be instructed in the technique as well as proper syringe and needle disposal, and be cautioned against reuse of these items.
Tuberculosis: As observed with other TNF blocking agents, tuberculosis associated with the administration of HUMIRA in clinical trials has been reported (see WARNINGS). While cases were observed at all doses, the incidence of tuberculosis reactivations was particularly increased at doses of HUMIRA that were higher than the recommended dose. All patients recovered after standard antimicrobial therapy. No deaths due to tuberculosis occurred during the clinical trials. Before initiation of therapy with HUMIRA, patients should be evaluated for active or latent tuberculosis infection with a tuberculin skin test. If latent infection is diagnosed, appropriate prophylaxis in accordance with the Centers for Disease Control and Prevention guidelines should be institucted to seek medical advice if signs/symptoms (e.g., persistent cough, wasting/weight loss, low grade fever) suggestive of a tuberculosis infection occur. Patients with Heart Failure: Cases of worsening congestive heart failure (CHF) and new onset CHF have been reported with TMF blockers. Cases of worsening the prophylaxis in carefully. Immunosuppression: The possibility exists for TNF blocking agents, including immunesuppression: The possibility exists for TNF blocking agents, including

64 patients with rheumatoid arthritis treated with HUMIRA, there was no evidence of depression of delayed-type hypersensitivity, depression of immunoglobulin levels, or change in enumeration of effector T- and B-cells and NK-cells, monocyte/macrophages, and neutrophils. The impact of treatment with HUMIRA on the development and course of malignancies, as well as active and/or chronic infections is not fully understood (see WARNINGS, ADVERSE REACTIONS, Intections and Malignancies). The safety and efficacy of HUMIRA in patients with immunosuppression have not been evaluated. Immunizations: No data are available on the effects of vaccination in patients receiving HUMIRA. Live vaccines should not be given concurrently with HUMIRA. No data are available on the secondary transmission of infection by live vaccines in patients receiving HUMIRA. Autoimmunity: Treatment with HUMIRA may result in the formation of autoanti-bodies and, rarely, in the development of a lupus-like syndrome. If a patient develops symptoms suggestive of a lupus-like syndrome following treatment with HUMIRA, treatment should be discontinued (see ADVERSE REACTIONS, Autoantibodies).

Methodrexare
HUMIRA has been studied in rheumatoid arthritis patients taking concomitant
MTX (see CLINICAL PHARMACOLOGY: Drug Interactions). The data do not suggest the need for dose adjustment of either HUMIRA or MTX.

gest the need for dose adjustment of either HUMIRA or MTX.
Anakinra
Concurrent administration of anakinra (an interleukin-1 antagonist) and another
TNF-blocking agent has been associated with an increased risk of serious infections, an increased risk of neutropenia and no additional benefit compared to
these medicinal products alone. Therefore, the combination of anakinra with other
TNF-blocking agents, including HUMIRA may also result in similar toxicities (see
WARNINGS, SERIOUS INFECTIONS).
Carcinogenesis, Mutagenesis, and Impairment of Fertility
Long-term animal studies of HUMIRA have not been conducted to evaluate the
carcinogenic potential or its effect on fertility. No clastogenic or mutagenic effects
of HUMIRA were observed in the in vivo mouse micronucleus test or the Salmonella-Escherichia coli (Ames) assay, respectively.
Pregnancy Pregnancy Category B - An embryo-fetal perinatal developmental toxicity study has been performed in cynomolgus monkeys at dosages up to 100
mg/kg (266 times human AUC when given 40 mg subcutaneous without MTX)
and has revealed no evidence of harm to the fetuses due to addiminumab. There
are, however, no adequate and well-controlled studies in pregnant women.
Because animal reproduction and developmental studies are not always predictive
of human response, HUMIRA (adalimumab) should be used during pregnancy
only if clearly needed.

of human response, HUMINA (auainiuman) should be accounted to only if clearly needed.

Pregnancy Registry: To monitor outcomes of pregnant women exposed to HUMIRA, a pregnancy registry has been established. Physicians are encouraged to register patients by calling 1-877-311-8972

Nursing Mothers: It is not known whether adalimumab is excreted in human milk or absorbed systemically after ingestion. Because many drugs and immunoglobulins are excreted in human milk, and because of the potential for serious adverse reactions in nursing infants from HUMIRA, a decision should be made whether to discontinue nursinn or to discontinue the drug, taking into account the imporreactions in musting initialist nion involved, a decision should be made whether in discontinue unising or to discontinue the drug, taking into account the importance of the drug to the mother.

Pediatric Use: Safety and effectiveness of HUMIRA in pediatric patients have not

been established.

Geriatric Use: A total of 519 patients 65 years of age and older, including 107 patients 75 years and older, received HUMIRA in clinical studies. No overall difference in effectiveness was observed between these subjects and younger subjects. The frequency of serious infection and malignancy among HUMIRA treated subjects over age 65 was higher than for those under age 65. Because there is a higher incidence of infections and malignancies in the elderly population in general, caution should be used when treating the elderly.

ADVERSE REACTIONS General: The most serious adverse reactions were (see WARNINGS):

**Serious Infections

* Neurologic Events

* Meurologic Events

The most common adverse reaction with HUMIRA was injection site reactions. In placebo-controlled trials, 20% of patients treated with HUMIRA developed injection site reactions (erythema and/or itching, hemorrhage, pain or swelling), compared to 14% of patients receiving placebo. Most injection site reactions were described as mild and generally did not necessitate drug discontinuation. The proportion of patients who discontinued treatment due to adverse events during the double-blind, placebo-controlled portion of Studies I, II, III and IV was 7% for patients taking HUMIRA and 4% for placebo-treated patients. The most common adverse events leading to discontinuation of HUMIRA were clinical flare reaction (0.7%), rash (0.3%) and pneumonia (0.3%). Because clinical trials are conducted under widely varying and controlled conditions, adverse reaction rates observed in clinical trials of a drug cannot be directly compared to rates in the clinical trials of another drug and may not predict the rates observed in a broader patient population in clinical practice.

Infections: In placebo-controlled trials, the rate of infection was 1 per patient-year in the HUMIRA-treated patients and 0.9 per patient-year in the placebo-treated patients. The infections consisted primarily of upper respiratory tract infections, bronothits and urinary tract infections. Most patients continued on HUMIRA after the infection resolved. The incidence of serious infections was 0.04 per patient-year in HUMIRA-treated patients and 0.02 per patient-year in placebo-treated patients. Serious infections beceived included penumonia, septic arthritis, prathetic and post-surgical infections, erysipelas, cellulitis, diverticulitis, and pyelonephritis (see WARNINGS).

Thirteen cases of tuberculosis including miliary, lymphatic, peritoneal, and pulmonary, were reported in clinical trials. Most of the cases of tuberculosis occurred within the first eight months after initiation of therapy and m

recrudescence of latent disease. Six cases of invasive opportunistic infections caused by histoplasma, aspergillus, and nocardia were also reported in clinical trials (see WARNINGS).

Malignancies: Among 2468 rheumatoid arthritis patients with moderately to severely active disease treated with HUMIRA in clinical trials for a mean of 24 months (4870 patient-years of therapy), 10 lymphomas were observed for a rate of 0.21 cases per 100 patient-years. This is approximately 5-fold higher than expected in an age- and sex-matched general population based on the Surveillance, Epidemiology, and End Results Database. Rates in clinical trials for HUMIRA can not be compared to rates of clinical trials of other TNF blockers and may not predict the rates observed in a broader patient population. (see WARNINGS: Malignancies). An increased rate of lymphoma has been reported in the rheumatoid arthritis patient population. Patients with rheumatoid arthritis, particularly those with highly active disease, are at a higher risk for the development of malignancies is not known. Thirty-eight malignancies, other than lymphoma, were observed. Of these, the most common malignancies were breast, colon, prostate, and uterine, which were similar in type and number to what would be expected in the general population.

Autoantibodies: in the controlled trials, 12% of patients treated with HUMIRA and 7% of placebo-treated patients that had negative baseline ANA fitters developed clinical signs suggestive of new-onset lupus-like syndrome. The patient improved following discontinuation of therapy. No patients developed lupus nephritis or central nervous system symptoms. The impact of long-term treatment with HUMIRA on the development of autoimmune diseases is unknown. Immunogenicity: Patients in Studies 1, II, and III were tested at multiple time

points for antibodies to adalimumab during the 6- to 12-month period. Approximately 5% (58 of 1062) of adult rheumatoid arthritis patients receiving HUMIRA developed low-titer antibodies to adalimumab at least once during treatment, which were neutralizing in vitro. Patients treated with concomitant MTX had a lower rate of antibody development than patients on HUMIRA monotherapy (1% versus 12%). No apparent correlation of antibody development to adverse events was observed. With monotherapy, patients receiving every other week dosing may develop antibodies more frequently than those receiving weekly dosing. In patients receiving the recommended dosage of 40 mg every other week as monotherapy, the ACR 20 response was lower among antibody-positive patients than among antibody-negative patients. The long-term immunogenicity of HUMIRA is unknown.

than among antibody-negative patients. The long-term immunogenicity of HUMIRA is unknown. The data reflect the percentage of patients whose test results were considered positive for antibodies to adalimumab in an ELISA assay, and are highly dependent on the sensitivity and specificity of the assay. Additionally the observed incidence of antibody positivity in an assay may be influenced by several factors including sample handling, timing of sample collection, concomitant medications, and underlying disease. For these reasons, comparison of the incidence of antibodies to adalimumab with the incidence of antibodies to other products may be mislaarlino.

booles to adaminated with the incontrol of authorities to adaminate of the insileading.

Other Adverse Reactions: The data described below reflect exposure to HUMIRA in 2468 patients, including 2073 exposed for 6 months, 1497 exposed for greater than one year and 1380 in adequate and well-controlled studies (Studies I, II, III, and IV), HUMIRA was studied primarily in placebo-controlled trials and in long-term follow up studies for up to 36 months duration. The population had a mean age of 54 years, 77% were female, 91% were Caucasian and had moderately to severely active rheumatoid arthritis. Most patients received 40 mg HUMIRA every other week?

severely active meditation arumus. most passents asserts the reverse other week.

Table 4 summarizes events reported at a rate of at least 5% in patients treated with HUMIRA 40 mg every other week compared to placebo and with an incidence higher than placebo. Adverse event rates in patients treated with HUMIRA 40 mg every other week. In Study III, the types and frequencies of adverse events in the second year open-label extension were similar to those observed in the one-year

double-blind portion.

Table 4: Adverse Events Reported by B5% of Patients Treated with HUMIRA During Placebo-Controlled Period of Rheumatoid Arthritis Studies d of Rheumatoid Arthritis
HUMIRA
40 mg subcutaneous
Every Other Week

	(N=/U5)	(N=690)
Adverse Event (Preferred Term)	Percentage	Percentage
Respiratory Upper respiratory infection Sinusitis Flu syndrome	17 11 7	13 9 6
Gastrointestinal Nausea Abdominal pain	9 7	8 4
Laboratory Tests* Laboratory test abnormal Hypercholesterolemia Hyperlipidemia Hematuria Alkaline phosphatase increased	8 6 7 5 5	7 4 5 4 3
Other Injection site pain Injection site pain Injection site rea Rash Accidental injury Injection site reaction** Back pain Urinary tract infection Hypertension	12 12 12 10 8 6 8	12 8 6 8 1 4 5

Unitary tact linection

Typerfension

Apparension

Laboratory test abnormalities were reported as adverse events in European trials

Laboratory test abnormalities were reported as adverse events in European trials

Den cinicule erythema and/or itching, hemorrhage, pain or swelling

Other Adverse Events

Other infrequent serious adverse events occurring at an incidence of less than 5% in patients treated with HUMIRA were:

Body As A Whole: Fever, infection, pain in extremity, pelvic pain, sepsis, surgery, thorax pain, tuberculosis reactivated

Cardiovascular System: Arrhythmia, atrial fibrillation, cardiovascular disorder, chest pain, congestive heart failure, coronary artery disorder, heart arrest, hypertensive encephalopathy, myocardial infarct, palpitation, pericardial effusion, pericarditis, syncope, tachycardia, vascular disorder

Collagen Disorder: Lupus erythematosus syndrome

Digestive System: Cholecystitis, cholelithiasis, esophagitis, gastroenteritis, gastrointestinal disorder, gastrointestinal hemorrhage, hepatic necrosis, vomiting

Endocrine System: Parathyroid disorder

Hemic And Lymphatic System: Apranulocytosis, granulocytopenia, leukopenia,

Endocrine System: Parathyroid disorder **Hemic And Lymphatic System:** Agranulocytosis, granulocytopenia, leukopenia, ymphoma like reaction, pancytopenia, polycythemia (see **WARNINGS, Hemato**

logic Events).

Metabolic And Nutritional Disorders: Dehydration, healing abnormal, ketosis, einemia, peripheral edema

—Skeletal System: Arthritis, bone disorder, bone fracture (not sponta-bone necrosis, joint disorder, muscle cramps, myasthenia, pyogenic synovitis, tendon disorder

artnnts, synovitis, tendon disorder Meoplasia: Adenoma, carcinomas such as breast, gastrointestinal, skin, urogen-ital, and others; lymphoma and melanoma. Nervous System: Confusion, multiple sclerosis, paresthesia, subdural hematoma, tremor

Respiratory System: Asthma, bronchospasm, dyspnea, lung disorder, lung func-

tion decreased, pleural effusion, pneumonia Skin And Appendages: Cellulitis, erysipelas, herpes zoster Special Senses: Cataract

Special Senses: Cataract
Thrombosis: Irrombosis leg
Urogenital System: Cystitis, kidney calculus, menstrual disorder, pyelonephritis
Adverse Reaction Information from Spontaneous Reports:
Adverse Reaction Information from Spontaneous Reports:
Adverse events have been reported during post-approval use of HUMIRA.
Because these events are reported voluntarily from a population of uncertain size,
it is not always possible to reliably estimate their frequency or establish a causal
relationship to HUMIRA exposure
Hematologic Events: Thrombocytopenia (see WARNINGS, Hematologic Events).
Hypersensitivity reactions: Anaphylaxis (see WARNINGS, Hypersensitivity
Reactions).

ns: cutaneous vasculitis

OVERDOSAGE
The maximum tolerated dose of HUMIRA has not been established in human. The maximum tolerated dose of HUMIHA has not open established in Indinans. Multiple doses up to 10 mg/kg have been administered to patients in clinical trials without evidence of dose-limiting toxicities. In case of overdosage, it is recommended that the patient be monitored for any signs or symptoms of adverse reactions or effects and appropriate symptomatic treatment instituted immediately.

Revised: July 2004 Ref. 03-5364-R6 U.S. Govt. Lic. No. 0043 05B-640-H547-1 MASTER

ABBOTT LABORATORIES